Author's response to reviews

Title: Complicated malaria and other severe febrile illness in a pediatric ward in Libreville, Gabon

Authors:

Marielle K Bouyou-Akotet (mariellebouyou@gmail.com)
Denise P Mawili-Mboumba (dpmawili@hotmail.com)
Eric Kendjo (eric.kendjo@gmail.com)
Ariane Eyang Ekouma (ary-eyang@yahoo.fr)
Omar Abdou Raouf (depparmyc.uss@gmail.com)
Edouard Engohan Allogho (depparmyc.uss@gmail.com)
Maryvonne Kombila (valenyine_favry@yahoo.fr)

Version: 2 Date: 6 January 2012

Author's response to reviews: see over
To the Editorial Office,

BMC Infectious Diseases (Section: Tropical and parasitological diseases)

Libreville, 6th January 2012

Dear Editors,

We would like to submit our revised manuscript according to the changes requested for peer review process.

All authors e-mail addresses have been added. Dr Engohan and Dr Raouf have the same office e-mail address.

All authors have seen and approved the revised manuscript.

We look forward to hearing from you.

Pr Bouyou-Akotet Marielle K.

Correspondence to: Marielle Karine Bouyou-Akotet: Department of Parasitology-Mycology-, Faculty of Medicine, Université des Sciences de la Santé, BP 4009 Libreville, Gabon. Tel/Fax: + 241 70 48 96. E-mail: mariellebouyou@gmail.com